Author's response to reviews

Title: Intraocular lens dislocation and tube shunt in the posterior chamber

Authors:

JAVIER MORENO-MONTAÑES MD, PhD (jmoreno@unav.es)
Alvaro Velázquez-Villoria MD (avelazquezvilloria@gmail.com)
Alfonso L. Sabater MD (alfonso.sabater@gmail.com)
Angel Salinas-Alamán MD, PhD (asalinas@unav.es)

Version: 4 Date: 28 April 2015

Author's response to reviews: see over
Pamplona, 28th April 15

Dear Editor,

I am sending you a 3nd versión of “Intraocular lens dislocation and tube shunt in the posterior chamber” with all the changes requested by the editor.

1. Clarify why consent was obtained from the patients relative, and not from the patient.

2. Requesting for Email-add

   Provide email address of each co-authors in the Title page

3. Modify Author’s Contribution.

This manuscript has been written according to the Instructions for Authors and it has not been published by any other publication.

Every author agree with the final writing of the manuscript and none of the authors has a financial, commercial or proprietary interest in any instrument mentioned in this manuscript.

Yours sincerely,

Javier Moreno-Montañés Phd, MD
Corresponding author
Phone: +34 948 296331
Fax: +34 948 296500
jmoreno@unav.es